Author’s response to reviews

Title: Impact of disease screening on awareness and management of hypertension and diabetes between 2011 and 2015: results from the China Health and Retirement Longitudinal Study

Authors:

Chihua Li (cl3106@cumc.columbia.edu)

L.H. Lumey (lumey@columbia.edu)

Version: 5 Date: 02 Apr 2019

Author’s response to reviews:

We thank the editor for suggestions and have updated our manuscript accordingly. Our specific responses are as follows:

1. Please amend your title so that it states that this study is a secondary analysis of data.

Response: As suggested, we changed the title to 'Impact of disease screening on awareness and management of hypertension and diabetes between 2011 and 2015: results from the China Health and Retirement Longitudinal Study'.